STATE OF MINNESOTA POLLUTION CONTROL AGENCY

In the Matter of the Proposed Rules Governing Waste Tire Dump Abatement STATEMENT OF NEED AND REASONABLENESS

INTRODUCTION

Improper waste tire storage and disposal threatens natural resources and the quality of the environment, and endangers the public health, safety and welfare. Waste tires provide an ideal breeding habitat for mosquitoes which carry and transmit the LaCrosse Encephalitis virus, which endangers young people. Tires also become a major fire hazard when stacked in piles. Tires do not start on fire easily, but once a tire pile begins to burn, it is almost impossible to extinguish. In addition to the routine hazards created by a fire, combustion reactions within a tire pile generate a run-off containing pyrolytic oil which when allowed to contaminate the ground water is a hazard to public health and the environment. Because of the problems associated with waste tire dumps stated above, the Legislature has declared tire dumps to be a nuisance. Minn. Stat. § 115A.906, subd. 1 (1984).

The Minnesota Pollution Control Agency (hereinafter "Agency") is proposing a set of rules to control the abatement of waste tire dumps. These proposed rules apply to waste tire dumps located within the State of Minnesota. The rules are proposed for adoption pursuant to the Agency's authority under Minn. Stat. §§ 116.07, subd. 4 (1984) and 115A.914, subd. 1 (1984).

The statement is divided into nine parts. After this introduction, part II provides an overview of the proposed rules. Part III discusses the legal and

historical background of the waste tire dump abatement rules. Part IV contains the Agency's explanation of the need for the proposed rules as a whole. Part V contains the Agency's explanation, part by part, of the reasonableness of the proposed rules. Part VI documents how the Agency has considered the methods for reducing the impact of the proposed rules on small businesses, pursuant to the requirements of Minn. Stat. § 14.115 (1984), "Small Business Considerations in Rulemaking." Pursuant to Minn. Stat. § 116.07, subd. 6 (1984), part VII documents the economic impacts of the proposed rules. Part VIII contains the Agency's conclusion. Part IX contains a list of exhibits relied on by the Agency to support the proposed rules. The exhibits are available for review at the Agency's offices at 1935 West County Road B2, Roseville, Minnesota 55113.

II. OVERVIEW OF PROPOSED RULES

Minn. Stat. § 115A.906, subd. 2 (1984) specifies a general waste tire dump abatement process. Minn. Stat. § 115A.906, subd. 2 (1984) states that:

. . . the agency may abate a nuisance by processing and removing the tires. Before taking any action to abate a nuisance, the agency shall give notice to the tire collector responsible for the nuisance that the tires to be processed and removed constitute a nuisance and demand that the tires be shredded or chipped or removed within a specified period. Failure of the tire collector to take the required action within the specified period shall result in the issuance of an agency order to abate the nuisance. The abatement order may include entering the property where the nuisance is located, taking tires into public custody, and providing for their processing and removal. The agency order may be enforced pursuant to the provisions of section 115.071.

Minn. Stat. § 115A.906, subd. 2 (1984).

Minn. Stat. § 115A.906, subd. 4 (1984) goes on to state in relevant part:
"The agency may reimburse a person or political subdivision for the costs of

-3abatement." Based on the general waste tire dump abatement process specified by statute above, the proposed rules provide the following administrative procedures for the abatement of waste tire dumps: A. A tire collector who owns or operates a tire dump is subject to Agency abatement action. As defined by Minn. Stat. § 115A.90, subd. 8, a tire collector is a person who owns or operates a site used for the storage, collection, or deposit of more than 50 waste tires. As defined by Minn. Stat. § 115A.90, subd. 9, a tire dump is an establishment, site, or place of business without a required tire collector or tire processor permit that is maintained, operated, used or allowed to be used for storing, keeping or depositing unprocessed waste tires. B. The Agency will issue a request for abatement action to the tire collector responsible for the tire dump. Requests for abatement actions will be issued to tire collectors responsible for tire dumps that meet the spending priorities established in Minn. Stat. § 115A.912, subd. 2. The availability of waste tire collection funds and the Agency's administrative ability to take

action will also be considered when the Agency moves to abate a tire dump

C. The request for abatement action will specify the action that must be

taken, the reasons the Agency requests the action, a time frame within which to

respond to the request, and the Agency's intended actions if the tire collector

D. The request for abatement action will request the tire collector

through the issuance of a request for abatement action.

does not comply with the request for abatement action.

responsible for the tire dump to submit an abatement plan to the Agency. This plan must propose a method for cleaning up the tire dump that is acceptable to the Agency. The plan must also propose a means for bringing the tire dump into compliance with technical standards specified in the proposed rules. The

- E. The response made by the tire collector to a request for abatement action must be incorporated into a stipulation agreement signed by the tire
- F. If the tire collector does not respond adequately to the request for abatement action or is in violation of the provisions of the stipulation agreement, the Agency may make a determination of inadequate response and issue the tire collector a tire dump abatement order. The order may provide for entering the property where the tire dump is located, taking tires into public custody, and arranging for their processing and removal.
- G. Tire collectors who notified the Agency of their existence under the emergency rules, who are requested by the Agency to abate, and who have entered into a stipulation agreement incorporating an abatement plan are eligible for a partial reimbursement of the costs incurred as a result of abatement action taken by the tire collector in accordance with a signed stipulation agreement. The proposed rules state the criteria that will be applied by the Agency for partial reimbursement.

In the proposed rules, the general procedures set out above have been divided into nine parts. These nine parts are as follows.

A. Pt. 7035.8000 is entitled "Scope." This part specifies the contents

-5and purpose of the proposed rules and the persons directly affected by them. B. Pt. 7035.8010 is entitled "Definitions." It contains definitions of specific terms used in the proposed waste tire dump abatement rules. C. Pt. 7035.8020 is entitled "Abatement Procedures." This part describes the criteria the Agency shall use in deciding which tire dumps to abate first, the procedures that the Agency shall follow in abating tire dumps, and the actions that responsible tire collectors must take to comply with an Agency request to abate. This part requires a tire collector issued a request for abatement action to submit an abatement plan to the Agency which states a proposal for cleaning up the tire dump. D. Pt. 7035.8030 is entitled "Contents of Abatement Plan." This part specifies what a responsible tire collector must include in an abatement plan to meet Agency approval. The abatement plan must provide for cleanup in staged "abatement increments" with completion within five years. This part also requires the responsible tire collector to enter into a stipulation agreement with the Agency incorporating the abatement plan. E. Pt. 7035.8040 is entitled "Inadequate Response to a Request for Abatement Action." This part specifies the four criteria the Agency will apply in determining whether a tire collector has failed to take the action requested by the Agency. If the Agency makes a determination of inadequate response, the Agency may issue a tire dump abatement order to the responsible tire collector. This part describes the tire dump abatement order. F. Pt. 7035.8050 is entitled "Technical Standards." This part sets out locational and fire hazard standards the tire collector must comply with during

-6abatement. The tire collector is required to remove all waste tires from areas where their presence would endanger human health due to the ideal mosquito breeding grounds tires provide when filled with water. Removal of tires from areas where they may become mosquito breeding grounds is critical because mosquitoes which transmit the LaCrosse Encephalitis virus have been found in Minnesota. The tire collector must reduce fire hazards at the tire dump by constructing fire lanes throughout the tire dump and by limiting the sizes of individual stockpiles contained in the tire dump. G. Pt. 7035.8060 is entitled "Operational Standards." This part sets out how the tire dump must be operated during abatement. Tire collectors must develop an emergency preparedness manual, and must report to the Agency on progress in abating the nuisance. This part also requires a tire collector to notify the director at the completion of each abatement increment, so the director may inspect the tire dump to certify the work has been completed in accordance with the abatement plan. H. Pt. 7035.8070 is entitled "Reimbursement." This part sets out the of abatement. This part specifies the costs eligible for reimbursement, the tire collectors eligible for reimbursement, and the maximum rate of reimbursement.

criteria and standards for partially reimbursing a tire collector for the costs

I. Pt. 7035.8080 is entitled "Reimbursement Disbursement." This part specifies the findings the Agency director must make before disbursing funds for reimbursement. This part also specifies that only the actual cost of abatement will be reimbursed, provided that those costs are not in excess of the rate established in the stipulation agreement.

III. LEGAL AND HISTORICAL BACKGROUND OF THE RULES

In 1969, the Minnesota Legislature directed the Agency to control solid waste disposal methods and practices and to adopt standards, regulations, and variances regarding solid waste. Minn. Laws 1969, ch. 1046, codified as Minn. Stat. § 116.01 et. seq. (1984).

The statutory authority of the Agency to adopt rules relative to the abatement of land pollution is found in Minn. Stat. § 116.07, subd. 4 (1984) (Rules and standards), which provides in relevant part:

Subd. 4. Rules and standards. . . .

Pursuant and subject to the provisions of chapter 14, and the provisions hereof, the pollution control agency may adopt, amend, and rescind rules and standards having the force of law relating to any purpose within the provisions of Laws 1969, chapter 1046, for the collection, transportation, storage, processing, and disposal of solid waste and the prevention, abatement, or control of water, air, and land pollution which may be related thereto, and the deposit in or on land of any other material that may tend to cause pollution. . . . Without limitation, rules or standards may relate to collection, transportation, processing, disposal, equipment, location, procedures, methods, systems or techniques or to any other matter relevant to the prevention, abatement or control of water, air, and land pollution which may be advised through the control of collection, transportation, processing, and disposal of solid waste . . . and the deposit in or on land of any other material that may tend to cause pollution . . .

Minn. Stat. § 116.07, subd. 4 (1984).

The Agency's directive to adopt rules for the administration of waste tire dump abatement is found in Minn. Stat. § 115A.914, subd. 1 (1984), which provides, ". . . The agency shall adopt rules for administration of waste tire collector and processor permits, waste tire nuisance abatement, and waste tire collection."

In 1985, the Minnesota Legislature directed the Agency to adopt emergency rules for the administration of waste tire dump abatement. Minn. Stat. § 115A.914 subd. 1, as amended in Minn. Laws 1985, First Special Session, ch. 13 § 232, provides:

Agency rules. The agency shall adopt rules for administration of waste tire collector and processor permits, waste tire nuisance abatement, and waste tire collection. Until December 31, 1985, the agency may adopt emergency rules for these purposes.

Minn. Laws 1985, First Special Session, ch. 13 § 232.

The Agency adopted emergency rules governing waste tire dump abatement on October 22, 1985. The emergency rules became effective on November 21, 1985. See Minn. Rules pts. 7035.8000 to 7035.8160 [Emergency] (1985).

Preparation of the proposed rules began in October 1984. At that time, Agency staff reviewed existing waste tire management programs throughout the United States to determine an appropriate direction for development of the Agency's waste tire dump abatement program. During this review, Agency staff prepared a questionnaire on waste tire management issues, including questions specifically dealing with waste tire dump abatement. See Appendix 1. The staff surveyed 49 state environmental protection agencies and the United States Environmental Protection Agency. Local governmental officials with experience in tire dump abatement were also contacted. A summary report was prepared to provide an overview of input. See Appendix 2.

On October 22, 1984 the Agency published a Notice of Intent to Solicit
Outside Opinion in the <u>State Register</u>. 9 S.R. 851. See Appendix 3. The notice
was mailed to the 87 county solid waste officers, 149 sanitary and demolition

landfill owners and operators, and 47 persons who requested to be placed on the Agency's mailing list for waste tire issues. The Agency staff sent a letter to this same group on October 19, 1984. This letter solicited information concerning the locations and relative sizes of waste tire dumps in the State. The letter included a request for information concerning regional, county, and/or local involvement in the area of waste tire management. See Appendix 4.

Considering both the general abatement process laid down by Minn. Stat. § 115A.906, subd. 2 and the information submitted to the Agency as a result of the Notice to Solicit Outside Opinion and the October 19, 1984 solicitation letter, the Agency staff prepared waste tire dump abatement issue statements. See Appendix 5. Based on the waste tire dump abatement issue statements, Agency staff began drafting the waste tire dump abatement rules.

On June 18, 1985, the Agency notified the interested parties on the waste tire program mailing list that the Board Solid and Hazardous Waste Committee would meet on June 24, 1985 to discuss issues associated with drafting waste tire dump abatement rules. At the June 24 committee meeting, staff presented, in general terms, the history of the promulgation of the rules, possible regulatory methods, and other issues related to waste tire dump abatement. See Appendix 6. Based on the staff's presentation and comments received from the public, the committee provided input and guidance on the development of the rules. The committee directed Agency staff to draft waste tire dump abatement rules for presentation to the committee in July 1985.

On July 22, 1985 Agency staff presented the draft rules to the Solid and Hazardous Waste Committee. At the meeting, Agency staff explained the contents

of the draft rules. See Appendix 7. Several comments were made at the meeting which required minor changes in the draft rules. However, no comments were made by the committee or the public which required a substantial change in the draft rules. The Agency Solid and Hazardous Waste Committee therefore approved the draft rules provided that Agency staff made the minor changes in the draft rules suggested by comments received at the meeting. The committee passed the draft rules on to the Agency Board for consideration at its August 1985 meeting. The committee directed Agency staff to conduct a public meeting in early August prior to the August Agency Board meeting to discuss the contents of the draft rules.

On August 5, 1985 the Agency notified the interested parties on the waste tire program mailing list that the Agency staff would present the draft rules at a public meeting held August 12, 1985. At the August 12 public meeting, Agency staff explained and discussed the contents of the draft waste tire abatement rules with interested parties.

On August 6, 1985 the Agency mailed a letter to the members of the Governor's Commission on Scrap Tires. The letter solicited comments from the members on the draft rules. See Appendix 8.

The Solid and Hazardous Waste Committee of the Agency Board conducted a meeting on August 26, 1985. At the meeting, Agency staff informed the committee of the concerns and comments that were raised at the August 12 public meeting, and addressed changes in the draft rules reflecting issues raised by the committee and interested parties at the July 22, 1985 committee meeting. See Appendix 9.

On August 27, 1985 the Agency Board gave Agency staff authorization to enter into the emergency rulemaking process with the proposed emergency rules governing waste tire dump abatement. See Appendix 10.

The proposed emergency rules were published in the <u>State Register</u> on September 23, 1985. 10 S.R. 687. See Appendix 11. Following the <u>State Register</u> publication date, a 25 day comment period was provided pursuant to emergency rulemaking procedures. Minn. Stat. § 14.30 (1984). The Agency received several comments during the comment period. See Appendix 12.

Considering the comments made during the 25 day comment period, the Agency Board conducted a meeting on October 22, 1985 and adopted the emergency rules governing waste tire dump abatement, as proposed, with one minor change. See Appendix 13. The Agency published a Notice of Adoption in the <u>State Register</u> on December 9, 1985. 10 S.R. 1311. See Appendix 14.

The emergency rules governing waste tire dump abatement became effective on November 21, 1985.

In working to develop permanent rules governing waste tire dump abatement, Agency staff considered the comments made by interested parties during the 25 day comment period provided prior to the Agency adopting the emergency rules. The Agency staff also considered problems with the abatement process provided in the emergency rules discovered as a result of implementing the emergency rules. The emergency rules provided for the following options.

A. Abatement Plan.

Under this option, a waste tire collector was required to submit an abatement plan to the Agency specifying how he/she would clean up the tire dump

while complying with certain abatement standards. Once approved by the Agency, the plan was to be incorporated into a stipulation agreement signed by the tire collector and the Agency. If the plan was followed, the Agency would provide for up to a \$10 per ton reimbursement of the costs of removal and processing.

B. Sign Over.

Under this option, the tire collector would sign the tire dump into public custody. The State would then provide for the cleanup of the tire dump. The State would seek recovery of the actual cost of abatement from the tire collector after abatement was completed.

During the 25 day comment period provided prior to the Agency adopting the emergency rules as well as at the Board meeting in which the Agency adopted the emergency rules, commenters expressed problems with the sign-over and reimbursement options provided for under the emergency rules. Staff also identified various problems associated with the sign-over option.

Agency staff presented these problems to the Agency Board Solid and Hazardous Waste Committee on May 19, 1986. See Appendix 15. Based on Agency staff's presentation and input from interested parties attending the meeting, the committee recommended that agency staff change the emergency rules by eliminating the sign-over option. The committee also directed Agency staff to investigate the claim that the reimbursement rate was too low and to establish a new reimbursement rate in the draft rules if needed.

Agency staff drafted the permanent rules based on the emergency rules, but excluding the sign-over option. Also, Agency staff also provided a new reimbursement rate.

On July 9, 1986 the Agency Board Solid and Hazardous Waste Committee conducted a meeting to discuss the draft permanent rules and the new reimbursement rate formula. See Appendix 16. The committee believed that while the formula adequately dealt with the costs associated with cleaning up passenger automobile and light truck tires, it did not adequately deal with costs associated with heavy truck and off-the-road vehicle tires. The committee therefore directed Agency staff to redraft the reimbursement rate formula to deal with oversize tires. The committee also recommended an upper limit be established on the processing cost portion of the formula to deal with potential processor monopoly problems.

Agency staff redrafted the reimbursement rate portion of the rules and presented it to the Agency Board Solid and Hazardous Waste Committee on July 21, 1986. See Appendix 17.

Following discussion, the committee recommended that the draft rules be presented at the Agency Board at its August 12, 1986 meeting for authorization to enter into the rulemaking process.

IV. NEED FOR THE PROPOSED RULES

Minn. Stat. § 14.23 (1984) requires an agency to make an affirmative presentation of facts establishing the need for and reasonableness of the rules proposed. In general terms, this means that an agency must prove that in enacting rules the agency is not being arbitrary or capricious. To the extent that need and reasonableness are separate, need has come to mean that a problem exists that requires administrative attention and reasonableness means that the

solution proposed by the agency is a proper one.

The proposed rules are needed to assist the Agency to abate land pollution, as required by Minn. Stat. § 116.07, subds. 2 and 4 (1984). The proposed rules are needed to make specific the abatement process laid down by Minn. Stat. § 115A.906 as required by Minn. Stat. § 115A.914, subd. 1. The rules are further needed to aid counties in developing plans for the abatement of waste tire dumps, as required by Minn. Stat. § 115A.914, subd. 2 (1984). The rules also respond to the legislative policy set out in Minn. Stat. § 116D.02, subd. 2 (m): to conserve natural resources by recycling.

A. Minn. Stat. § 116.07 (1984).

The Minnesota Legislature has required the Agency to adopt standards for,
"... the control of the collection, transportation, storage, processing, and
disposal of solid waste ... for the prevention and abatement of water, air,
and land pollution ... " Minn. Stat. § 116.07, subd. 2 (1984). The
Legislature has supplemented that basic provision and made it more specific with
the following:

Subd. 4. Rules and standards. . . .

Pursuant and subject to the provisions of chapter 14, and the provisions hereof, the pollution control agency may adopt . . . rules and standards having the force of law relating to any purpose within the provisions of Laws 1969, chapter 1046, for the collection, transportation, storage, processing and disposal of solid waste and the prevention, abatement, or control of water, air, and land pollution which may be related thereto, and the deposit in or on land of any other material that may tend to cause pollution. . . . Without limitation, rules or standards may relate to collection, transportation, processing, disposal, equipment, location, procedures, methods, systems or techniques or to any other matter relevant to the prevention, abatement or control of water, air, and land pollution which may be advised through the control of collection, transportation, processing, and disposal of solid waste . . . and the deposit in or on land of any other material that may tend to cause

pollution. . . .

Minn. Stat. § 116.07, subd. 4 (1984).

Waste tires are a solid waste. Moreover, waste tires tend to cause pollution and are a detriment to human health.

According to the Metropolitan Mosquito Control District and State health officials, a mosquito which carries and transmits the LaCrosse Encephalitis virus, which endangers young people, is present in Minnesota. This mosquito is generally found in wooded areas, and typically breeds in water filled stumps and holes in trees. However, stagnant water in whole tires also provides an ideal breeding habitat and is preferred to tree holes when available. See Appendix 18.

Tires become a major fire hazard when stacked in piles. Tires do not start on fire easily, but once a tire pile begins to burn, it is almost impossible to extinguish. In addition to the routine hazards created by a fire, combustion reactions within a tire pile generate a run-off containing pyrolytic oil (synthetic crude oil), gas, and carbon black. The generation of pyrolytic oil is a hazard to public health and the environment when allowed to contaminate the ground water.

In 1983, in the state of Virginia, a tire dump containing between 7 to 9 million waste tires caught on fire. The fire burned for over eight months. It cost the federal government 1.8 million dollars to fight the fire and clean up the area. Over 840,000 gallons of liquid containing pyrolytic oil were collected at the site.

The proposed rules for waste tire dump abatement will provide for the

processing, removal, and utilization of waste tires. The proposed rules will also decrease the risk of fire and mosquito infestation for tire dumps subject to abatement. Thus, the proposed rules will abate pollution of the land resulting from the improper disposal of waste tires, a solid waste.

B. Minn. Stat. § 115A.914 (1984).

The Minnesota Legislature has required the Agency to, "adopt rules for administration of waste tire collector and processor permits, waste tire nuisance abatement, and waste tire collection." Minn. Stat. § 115A.914, subd. 1 (1984).

The proposed rules are needed to control the abatement of waste tire dumps, which are by statute defined to be a nuisance. Minn. Stat. § 115A.906, subd. 1 (1984). The rules for waste tire dump abatement will establish a process through which the Agency will move to abate waste tire dump nuisances.

C. Minn. Stat. § 115A.906 (1984) (Statutory Abatement Process).

Minn. Stat. § 115A.906, subd. 1 (1984) establishes that, ". . . a tire dump unreasonably endangers the health, safety, and comfort of individuals and the public and is a nuisance." Minn. Stat. § 115A.906, subd. 2 (1984) goes on to state that:

. . . The agency may abate a nuisance by processing and removing the tires. Before taking any action to abate a nuisance, the agency shall give notice to the tire collector responsible for the nuisance that the tires to be processed and removed constitute a nuisance and demand that the tires be shredded or chipped or removed within a specified period. Failure of the tire collector to take the required action within the specified period shall result in the issuance of an agency order to abate the nuisance. The abatement order may include entering the property where the nuisance is located, taking tires into public custody, and providing for their processing and removal. . . .

Minn. Stat. § 115A.906, subd. 2 (1984).

The proposed rules make specific the abatement process laid down by Minn. Stat. § 115A.906, subd. 2 (1984). The proposed rules set out what is expected of a tire collector during the abatement process and the action the Agency will take in moving through the abatement process. The proposed rules are needed to provide a waste tire dump abatement process which is specific, and consistently used.

D. Minn. Stat. § 115A.914 (Guidance to Counties).

Minn. Stat. § 115A.914, subd. 2 (1984) provides that, ". . . Counties shall include collection and processing of waste tires in the solid waste management plan prepared under sections 115A.42 to 115A.46 "

Because the counties need to know how the Agency will move to clean up waste tire dumps in developing their waste tire management plans, the proposed rules are needed to aid the counties in developing these plans.

E. Minn. Stat. § 116D.02, Subd. 2(m).

The Minnesota Legislature, ". . . authorizes and directs that, to the fullest extent practicable the policies, regulations, and public laws of the state shall be interpreted and administered in accordance with the policies set forth in sections 116D.01 to 116D.06." Minn. Stat. § 116D.03, subd. 1 (1984).

Under Minn. Stat. § 116D.02 (1984) the Minnesota Legislature directs State government to "... use all practicable means ... to improve and coordinate state plans, functions, programs and resources to the end that the state may, ... (m) Conserve natural resources and minimize environmental impact by ... recycling materials to conserve both materials and energy. ... " Minn. Stat. § 116D.02, subd. 2(m), (1984).

Technological advances have provided a use for waste tires. Energy and valuable materials can be conserved by the reuse and recycling of waste tires. The proposed rules promote the conservation of valuable materials and energy resources as directed by the Legislature.

F. Summary.

The Agency believes the statutory authority set forth above establishes the need for rules governing the abatement of waste tire dumps. The proposed rules respond to that need.

V. REASONABLENESS OF THE PROPOSED RULES

A. Introduction.

The Agency is required to make an affirmative presentation of facts establishing the reasonableness of the proposed rules. Minn. Stat. § 14.23 (1984). Reasonableness is the opposite of arbitrariness and caprice and means that there is a rational basis for the Agency's proposed action. The purpose of this section is to demonstrate that each provision is a reasonable approach to its defined function.

B. Reasonableness of Proposed Abatement Approach.

In working to develop the waste tire dump abatement rules, the Agency considered various alternatives based on the general abatement process set down by Minn. Stat. § 115A.916, subd. 2 (1984). Subdivision 2 states that:

. . . The agency may abate a nuisance by processing and removing the tires. Before taking any action to abate a nuisance, the agency shall give notice to the tire collector responsible for the nuisance that the tires to be processed and removed constitute a nuisance and demand that the tires be shredded or chipped or removed within a specified period. Failure of the tire collector to take the required action within the specified period shall result in the issuance of an agency order to abate the nuisance. The abatement order may include entering

the property where the nuisance is located, taking tires into public custody, and providing for their processing and removal. The agency order may be enforced pursuant to the provisions of section 115.071.

Minn. Stat. § 115A.906, subd. 2 (1984).

The following four abatement process approaches were considered by the Agency:

1. Direct Agency abatement action.

Under this approach, the tire collector would be issued a request for abatement action. If the waste tire collector did not adequately respond to the request, an abatement order would be issued. The tires could then be taken into public custody and the State could provide for their cleanup. The attorney general could seek to recover all State abatement expenses from the responsible tire collector.

2. Stipulation agreement incorporating abatement plan.

Under this approach, waste tire collectors would be required to submit an abatement plan to the Agency specifying how they will clean up the tire dump while complying with certain abatement standards outlined by the Agency. Once approved by the Agency, the plan would be incorporated into a stipulation agreement signed by the tire collector and the Agency. If the tire collectors failed to submit a plan or sign a stipulation agreement or did not comply with the provisions of the stipulation agreement, the Agency could proceed to issue an abatement order, as in the first outlined abatement approach.

3. Sign over.

Under this approach, a tire collector would be allowed to sign the tire dump into public custody. The State would then provide for the cleanup of the

tire dump. The State would retain the right to seek recovery of funds for the actual cost of abatement.

 Partial reimbursement for cleanup activities taken by the tire collectors.

In reviewing the approach options, the Agency recognized that many tire collectors would not have the funds needed to clean up their tire dumps. Thus, the Agency also considered whether providing for partial reimbursement of cleanup costs would give tire collectors an incentive to actively clean up their tire dumps, reducing the ultimate cost to the State.

The Agency's reasons for accepting or rejecting the outlined approaches are discussed below.

1. Direct Agency abatement action.

This type of abatement process closely coincides with the abatement process outlined by Minn. Stat. § 115A.906, subd. 2 (1984). However, in researching waste tire dump abatement in other states, Agency staff discovered that when a direct enforcement approach was used, the tire dumps were set on fire shortly after the abatement notice was issued to the tire dump owners. Only one exception was found. In that case, the tire dump owner could not be located and the city took public custody of the tire dump. This tire dump was set on fire by vandals shortly thereafter. These sites are being tested to assess whether contamination caused by the oil residue emitted from the tires has made them hazardous. If the sites are found to be hazardous, the cost of cleanup would far outweigh the cost of processing and removing the waste tires.

Because of the arson problem, the Agency believes the direct abatement

action approach is not a reasonable approach in and by itself. While the Agency does propose to incorporate portions of this approach into the overall abatement process provided in the proposed rules, the direct abatement action approach is combined with incentives (negotiated abatement plans and partial reimbursement of costs) for tire dump owners or operators to actively clean up their tire dumps. It is hoped that these incentives will decrease the number of tire collectors who might use arson to escape liability.

2. Stipulation agreement incorporating an abatement plan.

This abatement process approach shares problems with the first approach. However, this approach provides an incentive to tire dump owners or operators which the first approach does not. Under this approach, the tire collector submits an abatement plan specifying how and when the tire dump will be cleaned up. The abatement plan must be incorporated into a stipulation agreement signed by the tire collector and the Agency. This approach provides the tire collector and the Agency with the discretion to negotiate an acceptable abatement plan. While this approach probably does not completely solve the arson problem associated with tire dump abatement, because it provides for a negotiated cleanup plan the process should be more acceptable to tire collectors, and they should be more willing to accept responsibility.

The Agency believes this approach is within the statutory authority of Minn. Stat. § 115A.906, subd. 2 (1984) and has incorporated it, along with partial reimbursement, into the proposed rules.

Sign over.

The Agency believes the sign-over option should be excluded from the rules

for two reasons. First, the Agency received comments from the public during the 25 day comment period provided prior to the Agency adopting the emergency abatement rules critical of the sign-over option, and in favor of direct Agency abatement action. It was thought that tire collectors with funds adequate to clean up their tire dumps would try to use sign-over in order to escape financial liability.

Second, the Agency discovered that a great deal of confusion existed regarding the recovery of expenses after sign-over. Many parties believed that since sign-over was cooperative, the State would not seek to recover abatement expenses. This was never the Agency's intent. Were this the case, no tire collector would ever agree to pursue cleanup himself or herself.

Considering the confusion and the comments, the Agency has deleted the sign-over option from the proposed rules.

 Partial reimbursement for cleanup activities taken by the tire collector.

While the Agency believes tire collectors owning or operating tire dumps are financially responsible for the cleanup of their tire dumps, the Agency believes that a partial reimbursement will lower the total cleanup costs by providing an incentive for tire collectors to clean up their tire dumps in the most cost efficient manner. The Agency believes this type of an incentive is needed to avoid the use of arson as a solution to tire dump liabilities.

The Agency believes that the most reasonable waste tire dump abatement approach combines the direct abatement, the stipulation agreement, and partial reimbursement approaches. The Agency proposes to oversee the cleanup of tire

dumps under a negotiated binding agreement signed by the responsible tire collector and the Agency, while providing an economic incentive to the tire collector. The proposed rules embody this approach.

C. Reasonableness of Rule as Drafted.

In the following pages, the reasonableness of each of the nine parts of the proposed rules are discussed.

1. Pt. 7035.8000 Scope.

This part, which gives an overview of the contents of the rule, is reasonable because it informs affected persons, the public, and other governmental units of the purpose and content of the rules.

2. Pt. 7035.8010 Definitions.

This part contains the definitions of key words and phrases used in the waste tire dump abatement rule. The definitions are needed so that the rule may be subject to consistent interpretation. Some of the definitions are identical to statutory definitions, and thus are not in need of additional justification. Other definitions require some explanation. The reasonableness of these definitions is set out below.

Subp. 2. Abate or Abatement. A definition of abatement is given to clarify what the Agency means when it requests a tire collector to take action to abate the tire dump. Abatement means to process and remove the tires in a manner accepted by the Agency. This definition is reasonable because the Agency does not wish to limit the kind of processing employed by the tire collector, yet wishes to ensure that the processing is environmentally acceptable. The Agency is concerned that tires from Minnesota sites not end up

polluting neighboring jurisdiction.

Because tire collectors are required to abate their tire dumps pursuant to a stipulation agreement, it is reasonable to define abatement as, "to process and remove the tires in a manner accepted by the Agency."

Subp. 3. Abatement increment. Because this term is unique to the waste tire dump abatement rule, it is necessary to define it. The definition establishes that an abatement increment is a period of time specified in an abatement plan, not greater than six months and not less than one month, during which a specified number of waste tires will be removed from the tire dump and processed. The period of time allowed for abatement increments is reasonable given the administrative and technical restrictions on the Agency's ability to evaluate progress in abating a tire dump.

Subp. 6. Flood plain. The definition of flood plain that is given should enable tire collectors to judge whether or not their dump is located in a flood plain, and thus subject to immersion. Because the definition can easily be applied, it is reasonable.

Subp. 7. Operator. The definition of operator is given to clarify who the operator of a tire dump is. The reference to the statutory definition of tire collector is provided to alert operators to their responsibility under the statute.

Subp. 8. Owner. The definition of owner is included to alert tire dump owners as to who the Agency will consider an "owner" and thus a responsible tire collector under the statute. As defined, a lessee can be an owner. This is reasonable because the lessee owns a possessory interest in the tire dump, and

is in actual possession of the tire dump.

Subp. 11. Ravine. The definition should provide tire collectors with an easily applied method of determining whether tires are located in a ravine, and thus subject to immersion. Because the definition can easily be applied, it is reasonable.

Subp. 12. Residuals from processing. The definition is needed to clarify a term unique to the waste tire dump abatement rule. Because the definition can easily be applied, it is reasonable.

Subp. 13. Responsible tire collector. The statute requires the Agency to issue requests to the "tire collector responsible for the nuisance." Minn. Stat. § 115A.906, subd. 2 (1984). The definition clarifies who the Agency will consider a tire collector responsible for the nuisance. The tire collector who is responsible must be an owner or operator, as defined in the rule. This is reasonable because the statute defines a tire collector as an owner or operator of a tire dump.

Subp. 14. Shoreland. The definition should provide tire collectors with an easily applied method of determining whether tires are located in a shoreland, and thus subject to immersion. Because the definition can easily be applied, it is reasonable.

Subp. 15. Sinkhole. The definition should provide tire collectors with an easily applied method for determining whether tires are located in a sinkhole, and thus subject to immersion. Because the definition can easily be applied, it is reasonable.

Subp. 18. Tire-derived products. A definition is provided because this

term is unique to the waste tire dump abatement rules. The definition is easily applied and extremely inclusive, and thus reasonable.

Subp. 22. Wetland. Rather than rely entirely on a technical definition of wetland, the reason for the use of the term in the rule was used to generate a more inclusive and easily applied definition. Tires that are filled with water can become mosquito breeding grounds. Thus, tires must be prevented from being covered with standing water. The definition of wetland given in the rule reflects this problem and thus is reasonable. The more technical definition was provided to assist tire collectors who might have knowledge of how their sites would be classified.

3. Pt. 7035.8020 Abatement Procedures.

This part sets out the procedures that the Agency will follow when moving to abate a tire dump, and the criteria that the Agency will use to decide which tire dumps to abate first.

Subp. 2. Abatement priorities. Subp. 2 of this part sets out the criteria that the Agency will use to establish abatement priority. The order of the criteria is derived from the spending priorities established by the Legislature in Minn. Stat. § 115A.912, subd. 2 (1984), and is therefore reasonable. In all cases, the Agency considers the number of waste tires in the tire dump. This is reasonable because risk increases with the number of the waste tires. In ranking tire dumps that pose fire hazards (item B) the Agency is required to consider both the resources that would be threatened by a fire at the tire dump, and the characteristics of the tire dump that increase the chance of fire at the tire dump. In ranking tire dumps that are located in densely populated areas

(item C), the Agency is required to consider the population that might be affected by fire or mosquito infestation of the tire dump, the susceptibility of the tire dump to fire or infestation, and the visual impact of the tire dump. In ranking the remaining tire dumps (item D), the Agency is required to consider the hazardous characteristics of a particular tire dump and the population and resources that are threatened by the tire dump. The criteria that the Agency will use to rank the tire dumps in items B through D are reasonable because they require the Agency to assess the risk of a tire dump both in terms of its potential impact on people and the environment and in terms of its potential to develop a hazardous condition such as a fire or mosquito infestation.

Subp. 3. Request for abatement action. Subp. 3 of this part outlines the content of the request for abatement action that will be issued to responsible tire collectors. The issuance of the document called in the rule the "request for abatement action" is required by Minn. Stat. § 115A.906, subd. 2 (1984). The statute requires that before the Agency can act to abate a tire dump nuisance, it must give the tire collector responsible for the nuisance notice that the tires to be processed and removed constitute a nuisance and demand that the tires be shredded or chipped or removed within a specified period. The document called the request for abatement action gives the responsible tire collector the required notice. As set out in the rule, the document is reasonable in that it will give responsible tire collectors adequate notice of the Agency's intentions. The notice will be in writing, set out the action that must be taken to comply, set out the time in which the tire collector must respond, the Agency's reasons for requesting the abatement, and the actions that

the Agency will take if the requested action is not taken in the requested time. At the time the request for abatement action is issued, the responsible tire collectors will have opportunity to come before the Agency to contest the Agency's right to issue the abatement request to them.

Subp. 4. Requested action. Subp. 4 of this part outlines the actions that will be requested in the requests for response action. Under this subpart, the responsible tire collector will be required to submit an abatement plan meeting criteria outlined in the rule. This requirement is reasonable in that it allows the tire collector to propose a plan for abating the tire dump (within the limits specified in the rule) and does not require the tire collector to capitulate to an Agency abatement scheme that might be burdensome. Under this subpart, the responsible tire collector is also required to agree to implement the abatement plan submitted to the Agency by signing a stipulation agreement with the Agency incorporating the plan. This is reasonable because the signing of such an agreement will provide the Agency and the tire collector with a clear and definite record of what will be done at the site. The signing of such an agreement should work to the benefit of both parties.

4. Pt. 7035.8030 Contents of Abatement Plan.

This part sets out the contents of the abatement plan that a responsible tire collector will be required to submit to comply with the request for abatement action.

Subp. 1. Goal. Subp. 1 of this part sets forth the goal of the Agency in pursuing abatement of the tire dump: to have the waste tires processed and removed. Minn. Stat. § 115A.906, subd. 2 (1984) requires that the Agency give

tire collectors notice that "the tires to be processed and removed" constitute a nuisance. Thus, requiring both processing and removal is justified by the statute. Minn. Stat. § 115A.906, subd. 2 (1984) also states that the Agency must demand of the tire collector that the tires in a tire dump be "shredded or chipped or removed within a specified period. . . ." Because the shredding or chipping of waste tires represents only a partial solution to the problems associated with the storage of large numbers of waste tires (only ending the mosquito hazard), the Agency believes that removal is necessary. Requiring removal alone is also not satisfactory in that removal merely shifts the problem to another location. Thus, the Agency believes that it is only reasonable to require responsible tire collectors both to process and remove the waste tires from the tire dump site.

Subp. 2. Processing On-site. Subp. 2 of this part specifies information that must be included in an abatement plan if a responsible tire collector wishes to process the waste tires at the tire dump site. The information required under this subpart is needed to enable the staff to assess whether the method of abatement proposed by the responsible tire collector will result in the abatement of the tire dump in an environmentally sound manner, and in accordance with Agency technical and operational standards contained in the rule at pts. 7035.8050 and 7035.8060.

The information required in items A through D is needed so that the staff can be assured that the method of processing proposed is technically feasible and environmentally sound, and that the tire derived products produced will have a user.

The information required under item E is reasonable because it is needed to allow the staff to ascertain that the tire dump will be abated in an expeditious manner. The maximum time period for abatement (five years) is reasonable because according to available staff information, no site should take longer than five years to be abated. Requiring the responsible tire collectors to establish abatement increments is reasonable because such increments will aid the Agency and the tire collector in verifying that the site is being abated on schedule.

The information required to be submitted under item F is needed so that the staff can evaluate progress in achieving compliance with the operational and technical standards contained in this rule. It is reasonable to require that the site be brought into compliance with the technical and operational rules within six months because this should, in the majority of cases, be easily achieved. The rule contemplates that there might be exceptions to this time period.

The information required to be submitted by item G is needed to enable the staff to be sure that the records that the responsible tire collector will keep will be adequate to allow the staff to verify that the required processing has been completed.

The information required to be submitted by item H is needed to assist the staff in establishing the reimbursement rate. It is reasonable to require the responsible tire collectors to submit this information as they are the most likely to have it.

Subp. 3. Processing off-site. Subp. 3 of this part specifies the

information that a responsible tire collector will be required to submit if processing the tires off-site.

Items A through C require the responsible tire collector to submit information that the Agency staff will use to assess whether the method of transporting and processing the waste tires is adequate. To make this evaluation, the staff needs to have the individuals who will be transporting the waste tires identified, and the equipment that will be used described. The staff also needs to know where and how the waste tires will ultimately be processed, to ensure that the proposed processing technique is environmentally sound and feasible.

The information that is required to be submitted under items D through G is needed for the reasons stated with regard to subp. 2, items E through H.

Subp. 4. Permitting during abatement. Subp. 4 of this part is needed to respond to those responsible tire collectors who wish to have their sites become permitted waste tire facilities. Requiring these responsible tire collectors to give the Agency notice of this intent is reasonable if the responsible tire collector wishes the Agency to accept an abatement plan designed to allow for continued operation. Requiring the responsible tire collectors to agree to develop a plan for bringing the site into compliance with the applicable rules is reasonable because continued operation of the tire dump site would not otherwise be allowable under the abatement action, and the Agency must have a mechanism for ensuring that permitted status is actually sought and achieved.

Subp. 5. New waste tires. Subp. 5 of this part allows responsible tire collectors to propose an abatement plan that allows for limited site operation.

It is reasonable to require the responsible tire collectors to develop a plan for the processing and removal of new waste tires because the abatement of the site would not otherwise be achieved, and the Agency must have a mechanism for ensuring that the new waste tires are eventually processed and removed.

5. Pt. 7035.8040 Inadequate Response to a Request for Abatement Action.

This part sets out the criteria that the Agency must use in deciding whether a responsible tire collector has failed to respond adequately to an Agency request to abate. It is reasonable to set forth these criteria in the rule to put responsible tire collectors on notice as to what will subject them to a determination of inadequate response and subsequent abatement order, and to narrow and make consistent the things that the Agency must find when deciding that a response has been inadequate.

The ground for the finding of inadequate response set forth in item A (failure to respond in time) is reasonable because the Agency cannot refrain from acting to abate a site for an unlimited period of time. The period of time in which response must be received will be set forth in the request for abatement action sent to the responsible tire collector or collectors. Minn. Stat. § 115A.906, subd. 2 (1984) requires that the Agency specify a period for the abatement action.

The ground for the finding of inadequate response set forth in item B (failure to submit an adequate abatement plan) is reasonable because under this rule, a tire collector may only proceed with abatement pursuant to a stipulation agreement incorporating the abatement plan. The reasonableness of the elements of the abatement plan was discussed above at page 28.

The ground for the finding of inadequate response set forth in item C (failure to sign stipulation agreement) is reasonable because the rule requires that the tire collector sign a stipulation agreement if he/she desires to comply with the Agency request. The reasonableness of the requirement that a stipulation agreement be signed was discussed above at page 28.

The ground for the finding of inadequate response set forth in item D (failure to comply with stipulation agreement) is reasonable because it gives the Agency the option to issue an abatement order, an administrative remedy which allows the Agency to proceed to abate a site, rather than proceeding to court to enforce the agreement.

Subp. 2. Abatement Order. Subp. 2 of this part notifies those who receive an Agency request to abate that failure to respond will subject them to direct Agency action to abate their tire dump. The actions that the Agency states that it will take are reasonable because they are the actions that the Agency is allowed to take under Minn. Stat. § 115A.906, subd. 2 (1985).

6. Pt. 7035.8050 Technical Standards.

Because a tire dump is a nuisance and an environmental hazard, it is reasonable to require that certain minimal standards be complied with during the abatement process. In general, the standards set forth in this section ensure that the danger posed by the tire dump from fires and mosquito infestation will be minimized during the abatement process.

Subp. 2. Indoor storage. Subp. 2 of this part is needed to minimize the hazards posed by the storage of waste tires indoors. The standard referenced in this part is used in all parts of the United States and is generally accepted by

fire protection agencies.

Subp. 3. Location of waste tire piles. The requirements set forth in subp. 3 of this part generally ensure that the waste tires will not be subject to immersion in water. If immersed in water, waste tires may become mosquito breeding grounds. Further, flood plains, sinkholes, shorelands, ravines, and wetlands are environmentally sensitive areas. It is therefore reasonable for the Agency to require a tire collector to remove tires located in these environmentally sensitive areas first.

Subp. 4. Dimensions of waste tire piles. Subp. 4 of this part sets forth standards that will ensure that the danger posed by a fire in the tire dump is minimized. Tire pile size limitations and the requirement that fire lanes be maintained are fire protection agency accepted methods of limiting the spread of fires, and thus are reasonable.

This subpart establishes that the tire collector shall construct fire lanes and waste tire stockpiles that meet the following requirements. Tire piles must have area not greater than 10,000 square feet and a vertical height not greater than 20 feet. A minimum 50-foot fire lane between the stockpiles must be created and maintained free of rubbish, equipment, and vegetation at all times. These requirements are reasonable because they are accepted in other jurisdictions and are needed to enable fire fighters to work efficiently.

Connecticut and New Hampshire have guidelines that tire stockpiles cannot exceed a 100×100 foot area. In addition, Connecticut requires a 50 foot fire lane between the piles, and restricts tire pile height to 20 feet.

Two other states which have guidelines for tire stockpiles are New York and

The Minnesota Fire Marshal's Office suggested that the use of both portable and stationary equipment to develop a "water curtain" between the piles is needed to absorb heat. A fifty foot fire lane is needed for this equipment. This action may help to contain a fire to a single pile. Although the area restriction is needed so that water needed to dissipate heat can reach all parts of the tire pile, a solid stream of water can be sprayed 100 feet. This does not dissipate much heat. The tire pile size proposed is reasonable because there is also equipment available that can reach 65 to 110 feet both vertically and horizontally.

The Minneapolis Fire Prevention Bureau investigates fire hazards as they become aware of them. The only requirement of the Minneapolis Fire Prevention Bureau is to have at least a 20 foot fire lane between tire piles. A 20 foot fire lane would allow enough space for emergency equipment to enter the area. However, the lane would not be wide enough to prevent the fire from spreading to adjacent piles because of the heat generated from the burning tires. Further, a 20 foot fire lane only allows for one direction of travel.

The height of the tire stockpile is definitely a factor when fighting a fire. If the stockpile is high, the fire will burn like a chimney and spread upward and outward, faster and hotter. The Minneapolis Fire Prevention Code, Article 173, section 27.203 restricts tire pile height to 20 feet. Tire fires are hard to fight because tires are basically waterproof. A high stockpile

further complicates matters because water cannot reach the tires to cool them down adequately, and the tires cannot be separated by hand-held hooks, as such hooks can only reach ten feet.

7. Pt. 7035.8060 Operational Standards.

Because a tire dump is a nuisance and an environmental hazard, it is reasonable to require that certain minimal standards be complied with during the abatement process. In general, the standards set forth in this section are designed to ensure that the tire dump is operated in an environmentally sound manner during the abatement process.

- Subp. 2. Burning. Subp. 2 of this part is needed to reduce the danger of fire at the tire dump. By restricting the use of open flames within ten feet of a tire pile, the danger of accidental fire should be reduced.
- Subp. 3. Emergency equipment. Subp. 3 of this part is needed to ensure that if a fire does begin at the tire dump, it can quickly be brought under control.
- Subp. 4. Emergency communications. Subp. 4 of this part is needed to ensure that, should a tire fire be started, local fire protection authorities can be contacted and will respond as needed.
- Subp. 5. Access. Subp. 5 of this part requires that a controlled access road be maintained. The access road must be maintained to ensure that emergency vehicles have access to the site; the road must be controlled to ensure than arsonists or illegal dumpers cannot reach the site when it is not in operation.
- Subp. 6. Vegetation. Subp. 6 of this part is needed to minimize fire danger at the tire dump site from potentially flammable dried vegetation.

Subp. 7. Storage limitation. Subp. 7 of this part is needed to ensure that the tires are accessible in the event of a fire, and that other potentially dangerous materials are not stored with the tires.

Subp. 8. Emergency manual. Subp. 8 of this part requires that an emergency manual be prepared. This manual is submitted along with the abatement plan. This manual is needed to ensure that the operating personnel know what to do and who to contact in the event that there is a fire or other emergency with potential off-site impacts at the tire dump. Yearly updating is required. This is reasonable because the dimension and operation of the tire dump will be changing as abatement proceeds. The director can also require an update. This is reasonable because the circumstances affecting the tire dump may change before a year has elapsed, and the director is in the best position to respond to new information affecting the tire dump.

Subp. 9. Emergency notification and reports. Subp. 9 of this part requires that the operator notify the director in the event of a fire or other emergency. This is reasonable because Agency personnel could assist the operator in responding to the emergency. This part also requires the operator to submit a report on the emergency to the director of the Agency. This report is needed to enable the director to request changes in the emergency manual.

Subp. 10. Operational record. Subp. 10 of this part requires that the operator maintain records of operations at the tire dump. The records required under this subpart are needed to enable the Agency staff to assess the progress of the abatement action.

Subp. 11. Inspection. Subp. 11 of this part is needed to ensure that the

Agency staff can assess the progress of the abatement action through inspection of the site.

8. Pt. 7035.8070 Reimbursement.

As stated above at page 20, in researching the various approaches to waste tire abatement employed in other jurisdictions, the staff discovered that when direct enforcement methods were used, arson resulted. When confronted with the cost of removing and processing waste tires, tire collectors preferred to torch the tire piles and risk an arson prosecution. This increased the ultimate cost of cleanup considerably, due to the contamination of the soil with the oily residue emitted from the burning tires.

Faced with the danger of increasing the environmental threat through direct enforcement actions, the Agency staff concluded that the money allocated for waste tire abatement should be used to reimburse tire collectors for the cost of abatement. The Agency has the authority to reimburse persons for the costs of abatement under Minn. Stat. § 115A.906, subd. 4 (1984). This part of the rule sets forth the criteria and limits on reimbursement of tire collectors for the costs of abatement.

Subp. 2. Eligibility of responsible tire collector. Subp. 2 of this part restricts eligibility for reimbursement. Only tire collectors who notified the Agency pursuant to the notice requirements contained in the emergency rules governing waste tire abatement can apply for reimbursement. This is reasonable because only those tire collectors who are in compliance with the law should receive reimbursement. Reimbursement is also restricted to tire collectors who have received an Agency request to abate and who have entered into an agreement

with the Agency incorporating an abatement plan. This is reasonable because abatement funds must be spent on priority sites (i.e. those that have been issued an abatement request) and it would only be reasonable to spend the abatement money on reimbursement that has been conducted in an acceptable manner, pursuant to an approved abatement plan.

Subp. 3. Eligible costs. Subp. 3 of this part restricts the cost that may be reimbursed to the cost of processing and removing waste tires accumulated before the effective date of the emergency rule. This is reasonable because after that date tire collectors were on notice that their sites would need to be abated, and they could have increased the cost of disposal to cover the cost of removal.

Subp. 4. Reimbursement rate. Subp. 4 of this part establishes the reimbursement rate. Essentially, this section of the rule seeks to force tire collectors to abate their dumps in the most cost-effective manner by basing the amount of reimbursement on the most cost-effective method of processing and removing the tires in a particular site. The reimbursement rate will be used to establish a limit on reimbursement. Under pt. 7035.8080, subp. 4, only actual cost will be reimbursed. The tire collector will not be reimbursed, however, for costs over the reimbursement rate.

In subp. 4, the rule sets out a formula that will be used to calculate what removal and processing should cost. For waste tires under a specified dimension (i.e. auto and light truck tires) the formula contains a constant for transportation cost per ton. This constant was derived from analysis presented in the study Scrap Tires in Minnesota. See Appendix 19. It is reasonable to

establish a constant for these tires because the cost of transporting these tires is known and unlikely to vary, and these tires constitute the vast majority of tires in tire piles. For waste tires over the specified dimension, the cost of transportation and processing will be established by the director on a case-by-case basis. The director will use information on the cost of transporting and processing these tires submitted by the tire collector. This is reasonable because the cost of transportation of these tires will vary from site to site and from tire type to tire type.

The cost of transporting the tires is added to the net cost of processing the tires per ton. The cost of processing the tires is established by the director, and will be based on the most cost-effective method of processing available. A limit of \$66 per ton is established in the rule. This limit on processing cost was derived from analysis presented in the study Scrap Tires in Minnesota, Appendix 19, and represents the cost of processing a ton of tires assuming no revenue is received for the tire derived product. It is reasonable to establish such a limit in the rule to account for processor monopolies.

The Agency believes the first assumption is a reasonable one. In the Scrap Tires in Minnesota study, the consultant used the figure of 1,000 passenger/light truck tires per load as one of the main assumptions. In making this assumption, the consultant did an analysis of the availability of transportation equipment in the State. The results indicated that 45-foot enclosed trailer vans were favored and in adequate supply. Based on his experience in the waste tire collection and waste tire dump abatement business,

the consultant indicated that, on the average, there are about 1,000 passenger/light truck tires per 45-foot enclosed trailer van. Agency staff discussed this assumption with waste tire collectors and trucking firms. They also indicated that adequate numbers of both 40 and 45 foot enclosed trailer vans are available for hire. They also indicated that 1,000 passenger/light truck tires per load is a realistic assumption for those trailer vans.

The Agency believes the second assumption is a reasonable one. In the study mentioned above, the consultant used an estimated trucking rate of \$1.25 per mile in analyzing collection service economics. The consultant contacted several trucking firms and the Minnesota Department of Transportation and determined that tires could be transported from any area in Minnesota at this rate. Staff's discussions with waste tire collectors and trucking firms confirm the consultant's figure.

The Agency believes the third assumption is also a reasonable one. The consultant, who has over 15 years experience developing operational systems for the collection, processing, and marketing of waste tires, indicates that there are approximately 100 waste tires (passenger/light truck tires) per ton.

A limit of \$66 per ton is established in the processing cost portion of the formula. This limit on processing costs was derived from analysis presented in the Scrap Tires in Minnesota study, and represents the cost of processing a ton of passenger/light truck tires assuming no revenue is received by the processing facility for the tire derived product. It is reasonable to establish such a limit in the rule to account for processor monopolies.

Subp. 5. Reimbursement total. After the total cost is established, the

rule requires the director to establish the total amount of money that will be spent on a particular site. In arriving at this amount, the director must consider the total amount of money available to the Agency for abatement of tire dumps and the statutory spending priorities. The total is then incorporated into the abatement stipulation, and may not be increased by more than ten percent. Requiring the director to consider the total amount of money available for reimbursement and the spending priorities established by the legislation is reasonable because should there be a shortage of funds, they must be spent as the Legislature intended. Limiting increases in the reimbursement total established is reasonable in that it will give tire collectors incentive to work within the funding limits established. Allowing the ten percent increase is also reasonable, in that it gives the Agency the flexibility to increase the level of reimbursement should initial estimates prove faulty.

9. Pt. 7035.8080 Reimbursement Disbursement.

This part of the rule limits the Agency's disbursement of funds to tire collectors. The limits set out in this part are needed to ensure that the money spent to reimburse tire collectors is actually spent on the cost of the cleanup.

Subp. 2. Findings. Subp. 2 of this part requires that the director determine (before disbursing money) that the portion of the cleanup for which reimbursement is sought was actually completed, that the abatement cost was actually incurred, and that the responsible tire collector is in compliance with all terms and conditions of the stipulation agreement. It is reasonable to require that the director determine that a portion of the cleanup has actually incurred, because this protects the Agency's abatement fund. It is reasonable

Subp. 3. Documentation. Subp. 3 of this part allows the director of the Agency to request additional documentation to make the determinations required in subp. 2. This is needed to ensure that the director has sufficient information to make the required determinations. Because of the importance of protecting the abatement funds, it is reasonable to give the director the power to request this information rather than relying on the information that the tire collector must submit under these rules and the stipulation agreement.

Subp. 4. Disbursement. Subp. 4 of this part establishes that only the actual cost of abatement will be reimbursed. Tire collectors will not receive more than the reimbursement rate and total established in the agreement, however. This is reasonable because it restricts the spending of abatement funds to the actual costs of abatement, and gives the tire collector an incentive to abate the tire dump in the most cost-effective manner.

VI. SMALL BUSINESS CONSIDERATIONS

Minn. Stat. § 14.115, subd. 2 (1984) requires State agencies proposing new rules which affect small businesses to consider the following methods for reducing the impact of the rules on small businesses:

^{. . . (}a) the establishment of less stringent compliance or reporting requirements for small businesses;

⁽b) the establishment of less stringent schedules or deadlines for compliance or reporting requirements for small businesses;

⁽c) the consolidation or simplification of compliance or

reporting requirements for small businesses:

(d) the establishment of performance standards for small businesses to replace design or operational standards required in the rule; and

(e) the exemption of small businesses from any or all requirements of the rule.

Minn. Stat. § 14.115, subd. 2 (1984).

The statute requires agencies to incorporate into proposed rules any of the methods listed in subd. 2 "that it finds to be feasible, unless doing so would be contrary to the statutory objectives that are the basis of the proposed rulemaking." Minn. Stat. § 14.115, subd. 3 (1984).

Minn. Stat. § 115A.906, subd. 1 (1984) states that, ". . . a tire dump unreasonably endangers the health, safety, and comfort of individuals and the public and is a nuisance." Minn. Stat. § 115A.906, subd. 2 (1984) goes on to state that: ". . . the Agency may abate a nuisance by processing and removing the tires." Minn. Stat. § 115A.90, subd. 9 (1984) defines a tire dump as, ". . . an establishment, site, or place of business without a required tire collector or tire processor permit that is maintained, operated, used or allowed to be used for storing, keeping, or depositing unprocessed waste tires. . . ." (Emphasis added.) Minn. Stat. § 115A.902, subd. 2 (1984) states that:

. . . A permit is not required for:

(1) a retail tire seller for the retail selling site if no more than 500 waste tires are kept on the business premises:

(2) an owner or operator of a tire retreading business for the business site if no more than 3,000 waste tires are kept on the business preimises;

(3) an owner or operator of a business who, in the ordinary course of business, removes tires from motor vehicles if no more than 500 waste tires are kept on the business premises;

(4) a permitted landfill operator with less than 10,000 waste tires stored above ground at the permitted site; or

(5) a person using waste tires for agricultural purposes if the

waste tires are kept on the site of use. Minn. Stat. § 115A.902, subd. 2 (1984).

The Agency may not abate a tire dump which is exempted from obtaining a permit from the Agency. All of the exemptions listed above pertain to small businesses and reduce the impact of the rules on these businesses. The Agency believes methods for reducing the impact of the rules on small businesses were provided for in Minn. Stat. § 115A.902, subd. 2 (1984).

The deadlines for compliance, the reporting requirements, and the operational standards specified in the proposed rules are general in nature. This allows a tire collector to propose the means for achieving these requirements. The Agency believes that by establishing these general requirements, the proposed rules address the concerns of small business to the maximum extent possible without undermining the goals of Minn. Stat. § 115A.906, subd. 2 (1984).

Section 14.115 assumes that if small businesses are affected by new rules, the impact will be negative. The law requires an agency to mitigate the negative impact if possible. While these proposed rules may have a negative impact, they also provide a substantial positive impact on small businesses. Both impacts will be described.

The proposed rules directly affect small businesses running tire dumps subject to abatement. Although abatement may increase operation costs, the proposed rules reduce the financial impact for tire collectors by providing a partial reimbursement for cleanup costs they will incur. If a tire collector can work efficiently, the actual costs incurred by the tire collector may be

The Agency actively sought and encouraged input from the regulated community, including affected businesses, during the drafting of the proposed rules. This activity was discussed in part III of this document. Many comments were received during this process from small businesses, and the rules were drafted to take many of those comments into account.

To reiterate, the Agency believes the proposed rules address the concerns of the small business to the maximum extent possible without undermining the goal of Minn. Stat. § 115A.906, subd. 2 (1984): to abate waste tire dumps.

VII. ECONOMIC CONSIDERATIONS

Minn. Stat. § 116.07, subd. 6 (1984) requires the Agency to consider economic factors in exercising its rulemaking process:

In exercising all its powers the pollution control agency shall give due consideration to the establishment, maintenance, operation and expansion of business, commerce, trade, industry, traffic, and other economic factors and other material matters affecting the feasibility

and practicability of any proposed action, including, but not limited to, the burden on a municipality of any tax which may result therefrom, and shall take or provide for such action as may be reasonable, feasible, and practical under the circumstances.

Minn. Stat. § 116.07, subd. 6 (1984).

As was discussed under part VI, "Small Business Considerations," the proposed rules may have a direct economic impact on businesses owning or running tire dumps subject to abatement action. However, the proposed rules alleviate the economic burden placed on businesses by providing partial reimbursement of the costs of abating the tire dumps.

Many tire collectors may choose to hire and contract for cleanup and transportation services in cleaning up their tire dumps. The proposed rules therefore encourage expansion of businesses involved in such activities. For waste tire dumps that the Agency abates, the Agency would contract for cleanup and transportation services, resulting in increased opportunities.

The proposed rules require that the waste tires be processed. The waste tire processing facilities receiving waste tires will benefit economically from the added influx of waste tires.

A municipality owning or operating a tire dump may be economically impacted by the proposed rules. However, the proposed rules provide a partial reimbursement for cleanup costs incurred by municipalities that own or operate a tire dump. The proposed rules therefore strive to alleviate the economic burden for municipalities owning or operating tire dumps.

The Agency believes the proposed rules address economic concerns to the maximum extent possible without undermining the goal of Minn. Stat. § 115A.906, subd. 2 (1984).

VIII. CONCLUSION

The Agency staff has in this document and its exhibits made its presentation of facts establishing the need for and reasonableness of the proposed waste tire dump abatement rules. This document constitutes the Agency's statement of need and reasonableness for the proposed waste tire dump abatement rules.

Date: 8-1-86

Thomas J. Kalitowski Executive Director

IX. LIST OF EXHIBITS

In drafting the proposed rules, the Agency relied on technical documents prepared by a number of sources. The following documents were used by Agency staff in developing these rules and are relied on by the Agency as further support for the reasonableness of the proposed rules. These documents are available for review at the Agency's Public Information Office at 1935 West County Road B2, Roseville, Minnesota 55113.

Agency Exhibit Number	<u>Title</u>
1	Waste Recovery, Inc. 1985. <u>Scrap Tires in Minnesota</u> . Minnesota Pollution Control Agency.
2	National Fire Protection Association. 1980. The Standard for Storage of Rubber Tires. NFPA 231D-1980 Edition.
3	Fish and Wildlife Service. 1979. <u>Classification of Wetlands</u> and Deep Water Habitats of the United States. U.S. Department of the Interior.

APPENDICES

Appendix Number	<u>Title</u>
1	Division of Solid and Hazardous Waste. 1984. Questionnaire on Waste Tire Management Issues. Minnesota Pollution Control Agency. Unpublished.
2	Solid and Hazardous Waste Division. 1984. "Report on Other State Activities Regarding Waste Tire Management." Minnesota Pollution Control Agency. Unpublished.
3	Notice of Intent to Solicit Outside Opinion. October 22, 1984 <u>State Register</u> . 9 S.R. 851. Responses.
	A. October 24, 1984 Letter from Itasca County B. November 12, 1984 Letter from Indianhead Truck Line, Inc.
4	Letter from Minnesota Pollution Control Agency to Interested Parties dated October 19, 1984, and Responses.
	A. October 22, 1984 Letter from Sherburne County B. October 25, 1984 Letter from Traverse County C. October 25, 1984 Letter from Martin County D. October 25, 1984 Letter from Scott County E. October 31, 1984 Letter from Washington County F. November 5, 1984 Letter from Polk County G. November 13, 1984 Letter from Becker County H. November 13, 1984 Letter from Rock County I. November 29, 1984 Letter from Goodhue-Wabasha Counties J. February 14, 1985 Letter from Anoka County K. March 15, 1985 Letter from Dakota County
5	Solid and Hazardous Waste Division. 1985. "Issue Statements for Waste Tire Abatement and Collection Rule Development." Minnesota Pollution Control Agency. Unpublished.
6	Memorandum from Minnesota Pollution Control Agency staff to members of Minnesota Pollution Control Agency Board Solid and Hazardous Waste Committee dated June 19, 1985.

	No.
7	Memorandum from Minnesota Pollution Control Agency staff to members of Minnesota Pollution Control Agency Board Solid and Hazardous Waste Committee dated July 18, 1985.
8	Letter from Minnesota Pollution Control Agency to Members of the Governor's Special Commission on Scrap Tires dated August 6, 1985.
9	Memorandum from Minnesota Pollution Control Agency staff to members of Minnesota Pollution Control Agency Board Solid and Hazardous Waste Committee dated August 22, 1985.
10	Minnesota Pollution Control Agency Board item dated August 27, 1985.
11	Proposed Emergency Rules Published in the September 23, 1985 State Register. 10 S.R. 687.
12	Comments Received by Minnesota Pollution Control Agency During 25 Day Comment Period.
	 A. October 15, 1985 Letter from Dakota County B. October 17, 1985 Letter from the Minnesota Chapter of the National Solid Waste Management Association C. October 18, 1985 Letter from Briggs and Morgan Law Offices D. October 18, 1985 Comments Submitted by Bernie Beermann
13	Minnesota Pollution Control Agency Board Item dated October 22, 1985.
14	Notice of Adoption in the December 9, 1985 <u>State Register</u> . 10 S.R. 1311.
15	Memorandum from Minnesota Pollution Control Agency staff to members of Minnesota Pollution Control Agency Board Solid and Hazardous Waste Committee dated May 15, 1986.
16	Memorandum from Minnesota Pollution Control Agency staff to members of Minnesota Pollution Control Agency Board Solid and Hazardous Waste Committee dated July 2, 1986.
17	Memorandum from Minnesota Pollution Control Agency Board staff to Minnesota Pollution Control Agency Board Solid and Hazardous Waste Committee dated July 18, 1986.
18	Division of Disease Prevention and Control. 1979. "The Association of Artificial Containers and LaCrosse Encephacitis Cases in Minnesota." Minnesota Department of Health. Published.
	Science Magazine. 1984. "The Tire Trap." Published.

19

Waste Recovery, Inc. 1985. <u>Scrap Tires in Minnesota</u>. pages 119-120, 128-133. Minnesota Pollution Control Agency. Published.